% EJC papers *must* begin with the following two lines.
\documentclass[12pt]{article}
\usepackage{e-jc}

% Please remove all other commands that change parameters such as
% margins or pagesizes.

% only use standard LaTeX packages
% only include packages that you actually need

% we recommend these ams packages
\usepackage{amsthm,amsmath,amssymb}

% we recommend the graphicx package for importing figures
\usepackage{graphicx}

% use this command to create hyperlinks (optional and recommended)
\usepackage[colorlinks=true,citecolor=black,linkcolor=black,urlcolor=blue]{hyperref}

% use these commands for typesetting doi and arXiv references in the bibliography
\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}

% all overfull boxes must be fixed;
% i.e. there must be no text protruding into the margins


% declare theorem-like environments
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

\newcommand{\qdp}{\hspace*{1.5mm}}%
\newcommand{\qqdp}{\hspace*{2.5mm}}
\newcommand{\xqdp}{\hspace*{5.0mm}}
\newcommand{\xxqdp}{\hspace*{10mm}}
\newcommand{\xquad}{\quad\qquad\quad}
\newcommand{\qdn}{\hspace*{-1.5mm}}
\newcommand{\qqdn}{\hspace*{-2.5mm}}
\newcommand{\xqdn}{\hspace*{-5.0mm}}
\newcommand{\xxqdn}{\hspace*{-10mm}}
\newcommand{\+}{&\qqdn}%
\newcommand{\ang}[1]{\langle{#1}\rangle}
\newcommand{\mb}[1]{\mathbb{#1}}
\newcommand{\mc}[1]{\mathcal{#1}}
\newcommand{\mf}{\mathbf}
\newcommand{\mk}{\mathfrak}
\newcommand{\fat}[3]{\left(#1;#2\right)_{#3}}
\newcommand{\ffat}[3]{\left[#1;#2\right]_{#3}}
\newcommand{\fnk}[3]{\left[\qdn\ba{#1}#2\\#3\ea\qdn\right]}
\newcommand{\ffnk}[4]{\left[\qdn\ba{#1}#3\\#4\ea{\!\Big|\:#2}\right]}
\newcommand{\fnkl}[4]{\left[{\qdn\ba{#1}#3\\#4\ea{#2}}\right.}
\newcommand{\fnkr}[4]{\left.{\ba{#1}#3\\#4\ea}{\Big|\:#2}\right]}
\newcommand{\dfnk}[6]{\left[\qdn\ba{#1}#2\\#3\ea{\!\!%
             \Big|\!\!}\ba{#4}#5\\#6\ea\qdn\right]}%%%
\newcommand{\binm}{\binom}
\newcommand{\sbnm}[2]{\Bigl(\!\ba{c}\!#1\!\\#2\ea\!\Bigr)}
\newcommand{\tbnm}[2]{\tbinom{#1}{#2}}
\newcommand{\zbnm}[2]{\genfrac{(}{)}{0mm}{2}{#1}{#2}}
\newcommand{\binq}[2]{{#1\brack#2}}
\newcommand{\sbnq}[2]{\Bigl[\!\ba{c}\!#1\!\\#2\ea\!\Bigr]}
\newcommand{\tbnq}[2]{\genfrac{[}{]}{0mm}{1}{#1}{#2}}
\newcommand{\zfrac}[2]{\genfrac{}{}{}{2}{#1}{#2}}
\newcommand{\double}[2]{\genfrac{}{}{0mm}{1}{#1}{#2}}
\newcommand{\doppio}[2]{\genfrac{}{}{0mm}{2}{#1}{#2}}
\newcommand{\triple}{\qdn\ba{c}{\wr}\\[-1mm]{\wr}\\[-1mm]{\wr}\ea\qdn}
\newcommand{\quint}{\mult{c}{:\\[-2.5mm]\vdots}}
\newcommand{\nnm}{\nonumber}
\newcommand{\be}{\begin{equation}}
\newcommand{\ee}{\end{equation}}
\newcommand{\ba}{\begin{array}}
\newcommand{\ea}{\end{array}}
\newcommand{\bmn}{\begin{eqnarray}}
\newcommand{\emn}{\end{eqnarray}}
\newcommand{\bnm}{\begin{eqnarray*}}
\newcommand{\enm}{\end{eqnarray*}}
\newcommand{\bln}{\begin{subequations}}
\newcommand{\eln}{\end{subequations}}
\newcommand{\blm}{\bln\bmn}
\newcommand{\elm}{\emn\eln}
\newcommand{\eqn}[1]{\begin{equation}#1\end{equation}}
\newcommand{\xalignx}[1]{\begin{align}#1%%%%%%%%%%%%%%
            \end{align}}     %%%align=auto-space%%%%%%
\newcommand{\xalignz}[1]{\begin{align*}#1%%%%%%%%%%%%%
            \end{align*}}    %%%align=auto-space%%%%%%
\newcommand{\xflalignx}[1]{\begin{flalign}#1%%%%%%%%%%
            \end{flalign}}   %%%flalign=max-space%%%%%
\newcommand{\xflalignz}[1]{\begin{flalign*}#1%%%%%%%%%
            \end{flalign*}}  %%flalign=max-space%%%%%%
\newcommand{\xalignatx}[2]{\begin{alignat}{#1}#2%%%%%%
            \end{alignat}}   %%%alignat=no-space%%%%%%
\newcommand{\xalignatz}[2]{\begin{alignat*}{#1}#2%%%%%
            \end{alignat*}}  %%%alignat=no-space%%%%%%
\newcommand{\xsplitx}[1]{\begin{split}#1\end{split}}%%
\newcommand{\xalignedx}[1]{\begin{aligned}#1%%%%%%%%%%
            \end{aligned}}  %%%aligned in equation%%%%
\newcommand{\xalignedatx}[2]{\begin{alignedat}{#1}#2%%
            \end{alignedat}}%%%alignedat in equation%%
\newcommand{\xgatherx}[1]{\begin{gather}#1%%%%%%%%%%%%
            \end{gather}}   %%%%%gather senza  & %%%%%
\newcommand{\xgatherz}[1]{\begin{gather*}#1%%%%%%%%%%%
           \end{gather*}}   %%%%%gather senza & %%%%%%
\newcommand{\xgatheredx}[1]{\begin{gathered}#1%%%%%%%%
            \end{gathered}}%gathered=no& in equation%%
\newcommand{\mult}[2]{\begin{array}{#1}#2\end{array}}%
\newcommand{\xmultx}[1]{\begin{multline}#1%%%%%%%%%%%%
            \end{multline}} %%%multline senza & %%%%%%
\newcommand{\xmultz}[1]{\begin{multline*}#1%%%%%%%%%%%
            \end{multline*}}%%%multline senza & %%%%%%
\newcommand{\alp}{\alpha}
\newcommand{\bet}{\beta}
\newcommand{\gam}{\gamma}

\newcommand{\lam}{\lambda}
\newcommand{\vrho}{\varrho}

\newcommand{\eps}{\epsilon}
\newcommand{\veps}{\varepsilon}
\newcommand{\del}{\delta}

\newcommand{\sig}{\sigma}
\newcommand{\vthe}{\vartheta}
\newcommand{\ome}{\omega}

\newcommand{\ii}{\imath}
\newcommand{\jj}{\jmath}
\newcommand{\kk}{\kappa}

\newcommand{\vsig}{\varsigma}
\newcommand{\vph}{\varphi}

\newcommand{\Del}{\Delta}
\newcommand{\The}{\Theta}
\newcommand{\Lam}{\Lambda}
\newcommand{\Ome}{\Omega}
\newcommand{\Sig}{\Sigma}

\newcommand{\Gam}{\Gamma}
\newcommand{\Ups}{\Upsilon}
\newcommand{\centro}[1]
           {\begin{center}#1\end{center}}
\newcommand{\sinistra}[1]
           {\begin{flushleft}#1\end{flushleft}}
\newcommand{\destra}[1]
           {\begin{flushright}#1\end{flushright}}
\newcommand{\centerbox}[2]{\centro{\begin{tabular}{|c|}\hline
\parbox{#1}{\mbox{}\\[2.5mm]#2\mbox{}\\}\\\hline\end{tabular}}}
\newcommand{\centrobmp}[3]{\vspace*{1mm}
        \centerbmp{#1}{#2}{#3}\vspace*{1mm}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% if needed include a line break (\\) at an appropriate place in the title

\title{\bf Rank and crank analogs for some colored partitions}

% input author, affilliation, address and support information as follows;
% the address should include the country, and does not have to include
% the street address

\author{Roberta R. Zhou
        \thanks{The first author's research was partially supported by
        %$^\ast$
%the scientific research foundation (XNB201503)
%of Northeastern University at Qinhuangdao,
the Fundamental Research Funds for the Central Universities (N142303009),
%and
the Natural Science Foundation of Hebei Province (A2015501066)
and NSFC(11501090).}\\
\small School of Mathematics and Statistics\\[-0.8ex]
\small Northeastern University at Qinhuangdao\\[-0.8ex]
\small Hebei 066004, P.~R.~China\\
\small\tt zhourui@neuq.edu.cn\\
\and  Wenlong Zhang
\thanks{The second author's research was partially supported by
the Fundamental Research Funds for the Central Universities.}\\
\small School of Mathematical Sciences\\[-0.8ex]
\small Dalian University of Technology\\[-0.8ex]
\small Dalian 116024, P. R. China\\
\small\tt wenlongzhang@dlut.edu.cn
}

% \date{\dateline{submission date}{acceptance date}\\
% \small Mathematics Subject Classifications: comma separated list of
% MSC codes available from http://www.ams.org/mathscinet/freeTools.html}

\date{%\dateline{Jan 1, 2012}{Jan 2, 2012}\\
\small  Mathematics Subject Classifications: Primary 11P83 and\\
Secondary 05A17, 05A19, 11F03, 11P81}

\begin{document}
\begin{thebibliography}{10}
\bibitem{and-86x} G.~E. Andrews.  
\newblock $q$-Series:
Their Development and Application in Analysis,
     Number Theory, Combinatorics, Physics, and Computer Algebra.
\newblock {\em CBMS Regional Conference Series in Mathematics,
     No.\:66; Amer. Math. Society}, 1986.

\bibitem{andrews-garvan-88} G.~E. Andrews and F.~G. Garvan.  
\newblock
Dyson's crank of a partition.  
\newblock {\em Bull. Amer. Math. Soc. (N.S.)},
18:167--171, 1988.

\bibitem{atkin.Swinnerton} A.~O.~L. Atkin and H.~P.~F. Swinnerton-Dyer.
\newblock Some properties of partitions.
\newblock {\em Proc. London Math. Soc.}, 4(3): 84--106, 1954.

\bibitem{bailey} W.~N. Bailey.
\newblock Generalized Hypergeometric Series.
\newblock {\em Cambridge University Press, Cambridge}, 1935.

\bibitem{b3r} B.~C. Berndt.
\newblock Ramanujan's Notebooks \emph{(Part~III)}.
\newblock {\em Springer-Verlag, New York}, pp.~510+xiv, 1991.

\bibitem{berndt-x}B.~C. Berndt.
\newblock Number Theory in the Spirit of Ramanujan.
\newblock {\em American Mathematical Society, Providence}, 2004.

\bibitem{berndt-ono} B.~C. Berndt and K. Ono.
\newblock Ramanujan's unpublished manuscript on the partition and tau functions.
\newblock {\em The Andrews Festschrift (Ed. D. Foata and G.-N. Han), Springer-Verlag, Berlin},
 39--110, 2001.

\bibitem{chan18rj} H.~C. Chan.
\newblock Ramanujan's cubic continued fraction and a generalization
      of his ``most beautiful identity".
\newblock {\em Int. J. Number Theory}, 6:673--680, 2010.

\bibitem{chu04b} W. Chu and L.~Di Claudio.
\newblock Classical Partition Identities and Basic Hypergeometric Series.
\newblock {\em Universit\`a degli Studi di Lecce, Edizioni del Grifo}, 2004.

\bibitem{dyson44} F.~J. Dyson.
\newblock Some guesses in the theory of partitions.
\newblock {\em Eureka (Cambridge)}, 8:10--15, 1944.

\bibitem{garvan88} F.~G. Garvan.
\newblock New combinatorial interpretations of Ramanujan's partition congruences
mod 5, 7, and 11.
\newblock {\em Trans. Amer. Math. Soc.}, 305:47--77, 1988.

\bibitem{garvan10} F.~G. Garvan.
\newblock Biranks for partitions into 2 colors.
\newblock {\em Ramanujan rediscovered}, 87--111,
      {\em Ramanujan Math. Soc. Lect. Notes Ser.}, 14,
      {\em Ramanujan Math. Soc.}, Mysore, 2010.

\bibitem{gas=90x} ~G. Gasper and ~M. Rahman.
\newblock Basic Hypergeometric Series \emph{(2nd ed.)}.
\newblock {\em Cambridge University Press, Cambridge}, 2004.

\bibitem{hammond-lewis} P. Hammond and R. Lewis.
\newblock Congruences in ordered pairs of partitions.
\newblock {\em Int. J. Math. Math. Sci.}, 47:2509--2512, 2004.

\bibitem{har=79x} G.~H. Hardy and E.~M. Wright.
\newblock An Introduction to the Theory of Numbers.
\newblock {\em The Clarendon Press, Oxford University Press, New York}, 1979.

\bibitem{kim} B. Kim.
\newblock A crank analog on a certain kind of partition function
     arising from the cubic continued fraction.
\newblock {\em Acta. Arith.}, 148:1--19, 2011.

\bibitem{kim11} B. Kim.
\newblock Partition statistics for cubic partition pairs.
\newblock {\em Electron. J. Combin.}, 18:$\#$P128, 2011.

\bibitem{lin14} B.~L.~S. Lin.
\newblock Arithmetic properties of overcubic partition pairs.
\newblock {\em Electron. J. Combin.}, 21(3):$\#$P3.35, 2014.

\bibitem{ramanujan} S. Ramanujan.
\newblock Some properties of $p(n)$, the number of partitions of $n$.
\newblock {\em Proc. Cambridge Philos. Soc.}, 19:207--210, 1919.

\bibitem{R1921mz} S. Ramanujan.
\newblock Congruence properties of partitions.
\newblock {\em Math. Zeit.}, 9:147--153, 1921.

\bibitem{toh} P.~C. Toh.
\newblock Ramanujan type identities and congruences for partition pairs.
\newblock {\em Discrete Mathematics}, 312:1244--1250, 2012.

\bibitem{winquist} L. Winquist.
\newblock An elementary proof of $p(11m+6)\equiv0 (mod 11)$.
\newblock {\em J. Combin. Theory}, 6:56--59, 1969.

\bibitem{zhang-wang} W. Zhang and C. Wang.
\newblock Ramanujan-type congruences for several partition functions.
\newblock {\em Int. J. Number Theory}, 10(3):641--652, 2014.

\bibitem{zhao-zhong} H. Zhao and Z. Zhong.
\newblock Ramanujan type congruences for a partition function.
\newblock {\em Electron. J. Combin.}, 18:$\#$P58, 2011.

\bibitem{zhou12} R. R. Zhou.
\newblock Multiranks for partitions into Multi-Colors.
\newblock {\em Electron. J. Combin.}, 19:$\#$P8, 2012.

%\bibitem{Bollobas} B{\'e}la Bollob{\'a}s.  \newblock Almost every
%  graph has reconstruction number three.  \newblock {\em J. Graph
%    Theory}, 14(1):1--4, 1990.
%
%\bibitem{WikipediaReconstruction} Wikipedia contributors.  \newblock
%  Reconstruction conjecture.  \newblock {\em Wikipedia, the free
%    encyclopedia}, 2011.
%
%\bibitem{FGH} J.~Fisher, R.~L. Graham, and F.~Harary.  \newblock A
%  simpler counterexample to the reconstruction conjecture for
%  denumerable graphs.  \newblock {\em J. Combinatorial Theory Ser. B},
%  12:203--204, 1972.
%
%\bibitem{HHRT} Edith Hemaspaandra, Lane~A. Hemaspaandra,
%  Stanis{\l}aw~P. Radziszowski, and Rahul Tripathi.  \newblock
%  Complexity results in graph reconstruction.  \newblock {\em Discrete
%    Appl. Math.}, 155(2):103--118, 2007.
%
%\bibitem{Kelly} Paul~J. Kelly.  \newblock A congruence theorem for
%  trees.  \newblock {\em Pacific J. Math.}, 7:961--968, 1957.
%
%\bibitem{KSU} Masashi Kiyomi, Toshiki Saitoh, and Ryuhei Uehara.
%  \newblock Reconstruction of interval graphs.  \newblock In {\em
%    Computing and combinatorics}, volume 5609 of {\em Lecture Notes in
%    Comput. Sci.}, pages 106--115. Springer, 2009.
%
%\bibitem{RM} S.~Ramachandran and S.~Monikandan.  \newblock Graph
%  reconstruction conjecture: reductions using complement, connectivity
%  and distance.  \newblock {\em Bull. Inst. Combin. Appl.},
%  56:103--108, 2009.
%
%\bibitem{RR} David Rivshin and Stanis{\l}aw~P. Radziszowski.
%  \newblock The vertex and edge graph reconstruction numbers of small
%  graphs.  \newblock {\em Australas. J. Combin.}, 45:175--188, 2009.
%
%\bibitem{Stockmeyer} Paul~K. Stockmeyer.  \newblock The falsity of the
%  reconstruction conjecture for tournaments.  \newblock {\em J. Graph
%    Theory}, 1(1):19--25, 1977.
%
%\bibitem{Ulam} S.~M. Ulam.  \newblock {\em A collection of
%    mathematical problems}.  \newblock Interscience Tracts in Pure and
%  Applied Mathematics, no. 8.  Interscience Publishers, New
%  York-London, 1960.

\end{thebibliography}


\maketitle

% E-JC papers must include an abstract. The abstract should consist of a
% succinct statement of background followed by a listing of the
% principal new results that are to be found in the paper. The abstract
% should be informative, clear, and as complete as possible. Phrases
% like "we investigate..." or "we study..." should be kept to a minimum
% in favor of "we prove that..."  or "we show that...".  Do not
% include equation numbers, unexpanded citations (such as "[23]"), or
% any other references to things in the paper that are not defined in
% the abstract. The abstract will be distributed without the rest of the
% paper so it must be entirely self-contained.

\begin{abstract}
  We establish some rank and crank analogs
for partitions into distinct colors and give combinatorial interpretations
for colored partitions such as partitions defined by Toh, Zhang and Wang
congruences modulo 5, 7.
  % keywords are optional

\bigskip\noindent \textbf{Keywords:} Partition congruences;
          rank analogs;
          Jacobi's triple product identity;
          Winquist's product identity
\end{abstract}

\section{Introduction and Motivation}

Let $p(n)$ be the number of unrestricted partitions of $n$, where
$n$ is nonnegative integer.
In 1921, Ramanujan~\cite{R1921mz} discovered the following
congruences \bnm
p(5n+4)&\equiv&0\pmod5\\
p(7n+5)&\equiv&0\pmod7.\enm
There exist many proofs in mathematical
literature,
%for which one can be found,
for example
%, in
~\cite{berndt-x,berndt-ono,%chu05a,ewell82,
ramanujan}.
%watson1938}.

In 1944, F.~J.~Dyson~\cite{dyson44} defined the rank of a partition
as the largest part minus the number of parts. Let $N(m,n)$ denote
the number of partitions of $n$ with rank $m$ and let $N(m,t,n)$
denote the number of partitions of $n$ with rank congruent to $m$
modulo $t$. In 1953, A.~O.~L. Atkin and H.~P.~F.
Swinnerton-Dyer~\cite{atkin.Swinnerton} proved
\[N(0,5,5n+4)=N(1,5,5n+4)=\cdots=N(4,5,5n+4)=\frac{p(5n+4)}5\]
and
\[N(0,7,7n+5)=N(1,7,7n+5)=\cdots=N(6,7,7n+5)=\frac{p(7n+5)}7.\]
Following from the fact that the operation of conjugation reverses
the sign of the rank, the trivial consequences are
\[N(m,n)=N(-m,n)\quad\text{and}\quad N(m,t,n)=N(t-m,t,n).\]

Hammond and Lewis~\cite{hammond-lewis} %investigated some elementary
%results for 2-colored partitions mod 5. They
defined birank
and
explained that the residue of the birank mod $5$ divided %the
2-colored partitions of $n$ into $5$ equal classes provided
$n\equiv2,3\:\:\text{or}\:\:4\pmod{5}$. F. G. Garvan~\cite{garvan10}
found two other analogs the Dyson-birank and the 5-core-birank.

In 2010, Chan~\cite{chan18rj} introduced the cubic partition
$a(n)$ as the number of 2-color partitions
of n with colors red and blue subjecting to the restriction that the color blue appears
only in even parts, and obtained the
following congruence
\[a(3n+2)\equiv0\pmod3.\]
Another proof has been given by B. Kim~\cite{kim}. He
defined a crank analog $M'(m,N,n)$ for $a(n)$ and proved that
\[M'(0,3,3n+2)\equiv M'(1,3,3n+2)\equiv M'(2,3,3n+2)\pmod{3},\]
for all nonnegative integers $n$, where $M'(m,N,n)$ is the number of
partition of $n$ with crank congruent to $m$ modulo $N$.
Later, B. Kim~\cite{kim11} gave two partition statistics which explained the
partition congruences about cubic partition pairs $b(n)$.
Here, $b(n)$ is the number of 4-color partitions of $n$
with colors red, yellow, orange, and blue subjecting
to the restriction that the colors orange and blue appear only in even parts.

About further research of arithmetic properties of cubic partitions,
overcubic partitions and other colored partitions,
some interesting results can be found in
~\cite{lin14,toh,zhang-wang,zhao-zhong}.
The first author~\cite{zhou12} of the present paper generalized Hammond-Lewis birank
and gave combinatorial interpretations for some colored partitions.

The paper is organized as follows. In Section 2, we introduce necessary notation and
some preliminary results. In Section 3, we aim to provide two partition statistics
for two colored partition congruences modulo 5. We establish six rank or crank analogs
for six colored partition with modulus 7 and give combinatorial interpretations
in Section 4.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Preliminary results}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
For the two indeterminates $q$ and $z$ with $|q|<1$, the $q$-shifted
factorial of infinite order %and the modified Jacobi theta function are
is defined %respectively
by
\[(z;q)_\infty=\prod_{n=0}^\infty(1-zq^n)
%\quad\text{and}\quad \ang{z;q}_\infty=(z;q)_\infty(q/z;q)_\infty
\]
where the multi-parameter expression for the former will be
abbreviated as
\[\ffat{\alp,\bet,\cdots,\gam}{q}{\infty}
=(\alp;q)_\infty(\bet;q)_\infty \cdots(\gam;q)_\infty.\]
%For brevity we denote $E_m%=\prod_{k=1}^\infty(1-q^{km})
%=(q^m;q^m)_\infty$.

The main purpose of this paper is to define rank and crank analogs for partition
into colors and prove colored partition congruences applying the
method of~\cite{garvan88}, which uses roots of unity.
Jacobi triple product identity,
the modified Jacobi triple product identity and Winquist
product identity
are given as follows:
\begin{itemize}
\item Jacobi triple product identity%~\cite{jacobi}
~\cite{and-86x,bailey,b3r,%ewell81,
gas=90x,har=79x%,lewis84
}:
\eqn{\sum_{n=-\infty}^{+\infty} (-1)^{n}\:q^{\zbnm{n}{2}}\:x^{n}
\:=\:\label{jacobi} \ffat{q,x,q/x}{q}{\infty}.}

\item Modified Jacobi triple product identity~\cite{hammond-lewis}:
\eqn{\ffat{q,zq,q/z}{q}{\infty}
=\label{modi-Jacobi}\sum_{n\geq0}(-1)^n(z^n+z^{n-1}
    +\cdots+z^{-n})q^{\zbnm{n+1}{2}}.}
\item Winquist product identity~\cite{chu04b,winquist}:
\bmn\xxqdn
\+\+\nnm(q;q)_{\infty}^4\ffat{x,q/x}{q}{\infty}^2\ffat{x^2,q/{x^2}}{q}{\infty}\\
\+=\+\label{winq}\sum_{i,j={-\infty}}^{+\infty}
(-1)^{i+j}q^{3\zbnm{i}{2}+3\zbnm{j}{2}+j}(1-3i+3j)
\{x^{3i+3j}-x^{4-3i-3j}\}. \emn
\end{itemize}

By replacing $q$ by $q^2$ in \eqref{winq},
splitting into two bilateral sums on right hand side of the resulting equation,
and replacing $j\to j-1$ in the first double sum,
and $i\to i+1$ in the second double sum,
the resulting
formula can be transformed as
\begin{itemize}
\item Modified Winquist product identity
\bmn\xxqdn
\+\+\nnm(q^2;q^2)_{\infty}^4\ffat{x,q^2/x}{q^2}{\infty}^2
\ffat{x^2,q^2/{x^2}}{q^2}{\infty}\\
\+=\+\label{modify-winq}\sum_{i,j={-\infty}}^{+\infty}
(-1)^{i+j}q^{6\zbnm{i}{2}+6\zbnm{j}{2}+3i-j+1}(2+3i-3j)
\{(x/q)^{3i+3j-3}-(x/q)^{1-3i-3j}\}. \emn
\end{itemize}

Dividing both sides by $1+x$ in~\eqref{winq}
and applying L'H\^opital's rule for the limit $x\to -1$,
we have
\bmn\xxqdn
(q;q)_{\infty}^2(q^2;q^2)_{\infty}^4
=\label{winq-1}\sum_{i,j={-\infty}}^{+\infty}
q^{3\zbnm{i}{2}+3\zbnm{j}{2}+j}
\frac{(1-3i+3j)(2-3i-3j)}{4}. \emn

Divide both sides by $1-q^2/{x^2}$ in~\eqref{modify-winq}
and utilize L'H\^opital's rule for the limit $x\to q$
to obtain
\bmn\xxqdn
(q;q)_{\infty}^4(q^2;q^2)_{\infty}^2
=\label{modify-winqq}\sum_{i,j={-\infty}}^{+\infty}
(-1)^{i+j}q^{6\zbnm{i}{2}+6\zbnm{j}{2}+3i-j+1}
(2+3i-3j)(3i+3j-2). \emn

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

After Andrews and Garvan~\cite{andrews-garvan-88}, for a partition
$\lam$, we define $\#(\lam)$ is the number of parts in $\lam$ and
$\sigma(\lam)$ is the sum of the parts of $\lam$ with the convention
$\#(\lam)=\sigma(\lam)=0$ for the empty partition $\lam$. Let
$\mc{P}$ be the set of all ordinary partitions,
%$\mc{D}$ be the set of all partitions into distinct parts,
%$\mc{O}$ be the set of all partitions into odd parts,
$\mc{D}\mc{O}$ be the set of all
partitions into distinct odd parts.

For a given partition
$\lam$, the crank $c(\lam)$ of a partition is defined as
\begin{equation*}
c(\lam):=\begin{cases}
\ell(\lam),&\text{if}\:r=0;\\
\omega(\lam)-r,&\text{if}\:r\geq1,
\end{cases}
\end{equation*}
where $r$ is the number of 1's in $\lam$,
$\omega(\lam)$ is the number of parts in $\lam$ that are strictly larger
than $r$ and $\ell(\lam)$ is the largest part in $\lam$.
By extending the set of partitions $\mc{P}$
to a new set $\mc{P^*}$ by adding two additional copies of
the partition $1$, say $1^*$ and $1^{**}$, B. Kim~\cite{kim,kim11} obtains
\begin{equation*}
\frac{(q;q)_\infty}{\ffat{zq,z^{-1}q}{q}{\infty}}
=\sum_{\lam\in \mc{P^*}}wt(\lam)z^{c^*(\lam)}q^{\sigma^*(\lam)},
\end{equation*}
where $wt(\lam)$, $c^*(\lam)$, and $\sigma^*(\lam)$
are defined as follows.
Denote the weight $wt(\lam)$ for $\lam\in \mc{P^*}$ by
\begin{equation*}
wt(\lam):=\begin{cases}
1,&\text{if}\:\lam\in \mc{P}, \lam=1^*, \text{or}\: \lam=1^{**};\\
-1,&\text{if}\:\lam=1,
\end{cases}
\end{equation*}
and denote the extended crank $c^*(\lam)$ by
\begin{equation*}
c^*(\lam):=\begin{cases}
c(\lam),&\text{if}\:\lam\in \mc{P};\\
0,&\text{if}\:\lam=1;\\
1,&\text{if}\:\lam=1^*;\\
-1,&\text{if}\:\lam=1^{**}.
\end{cases}
\end{equation*}
Finally, denote the extended sum parts function
$\sigma^*(\lam)$ in the following way:
\begin{equation*}
\sigma^*(\lam):=\begin{cases}
\sigma(\lam),&\text{if}\:\lam\in \mc{P};\\
1,&\text{otherwise}.
\end{cases}
\end{equation*}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Rank analogs for colored partitions congruences modulo 5}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In this section, we establish two statistics that divide the relevant
partitions into equinumerous classes and present the combinatorial
interpretation for colored partition congruences modulo 5.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We denote
\[C_{1^23^2}=\{(\lam_1,\lam_2,3\lam_3,3\lam_4)|
\lam_1,\lam_2,\lam_3,\lam_4\in\mc{P}\}.\] For $\lam\in
C_{\!1^2\!3^2\!}$, we define the sum of parts $s_{\!1^2\!3^2\!}(\lam)$,
and rank analog
$r_{\!1^2\!3^2\!}(\!\lam)$ by
\bnm
s_{\!1^2\!3^2\!}(\lam)&=&\sigma(\lam_1)+\sigma(\lam_2)+3\sigma(\lam_3)+3\sigma(\lam_4)\\
r_{1^23^2}(\lam)&=&\#(\lam_1)-\#(\lam_2)+\#(\lam_3)-\#(\lam_4).
\enm
The number of 4-colored partitions of $n$ if $s_{\!1^2\!3^2\!}(\lam)\!=\!n$
having $r_{\!1^2\!3^2\!}(\!\lam)\!=\!m$ will be written as
$N_{\!C_{\!1^2\!3^2}\!}(m,n)$, and $N_{\!C_{1^2\!3^2}\!}(m,t,n)$
is the number of such 4-colored partitions of $n$ having rank analog
$r_{\!1^2\!3^2\!}(\lam)\equiv m\pmod t$. Now, summing over all
4-colored partitions $\!\lam\in C_{1^2\!3^2}$ gives
\[N_{\!C_{\!1^2\!3^2}\!}(m,n)
\!=\!\sum_{\double{\lam\in
C_{1^23^2},s_{\!1^2\!3^2\!}(\lam)=n,}{r_{1^23^2}(\lam)=m}}1.\]

Since
\[r_{1^23^2}(\lam_1,\lam_2,\lam_3,\lam_4)
=-r_{1^23^2}(\lam_2,\lam_1,\lam_4,\lam_3),\] hence
\[N_{\!C_{1^23^2}\!}(m,n)
\!=\!N_{\!C_{1^23^2}\!}(\!-\!m,n) \quad\text{and}\quad
N_{\!C_{1^23^2}\!}(m,t,n)
\!=\!N_{\!C_{1^23^2}\!}(t\!-\!m,t,n).\]
Then we have
\eqn{\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{C_{1^23^2}}(m,n)z^mq^n
=\label{NC1^23^2gf}
\frac{1}
{(zq;q)_\infty(z^{-1}q;q)_\infty(zq^3;q^3)_\infty(z^{-1}q^3;q^3)_\infty}.}
By putting $z=1$ in the identity~\eqref{NC1^23^2gf}, we find
\[\sum_{m=-\infty}^{\infty}N_{C_{1^23^2}}(m,n)=c(n),\]
where $c(n)$ is defined by
$\sum_{n=0}^\infty c(n)q^n=\frac1{(q;q)^2_\infty(q^3;q^3)^2_\infty}$.

\begin{theorem}\label{c}
For $n\geq0$,
\[N_{\!C_{1^23^2}\!}(0,5,5n+2)\!=\!N_{\!C_{1^23^2}\!}(1,5,5n+2)
\!=\!N_{\!C_{1^23^2}\!}(2,5,5n+2)
\!=\!\frac{c(5n\!+\!2)}5.\]
\end{theorem}
It can also prove the identity in Zhang and Wang~\cite{zhang-wang}:
$c(5n+2)\equiv0\pmod5$.
\begin{proof}
Suppose $\zeta$ is primitive $5th$ root of unity. By setting $z=\zeta$
in \eqref{NC1^23^2gf},
we write
\bnm
\+\+\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{C_{1^23^2}}(m,n)\zeta^mq^n
=\frac{1}{\ffat{\zeta q,\zeta^{-1}q}{q}{\infty}
      \ffat{\zeta q^3,\zeta^{-1}q^3}{q^3}{\infty}}\\
\+=\+\frac{1}
        {(q^5;q^{5})_\infty(q^{15};q^{15})_\infty}
        \times\ffat{q,\zeta^2 q, \zeta^{-2}q}{q}{\infty}
            \ffat{q^3,\zeta^2q^3,\zeta^{-2}q^3}{q^3}{\infty}. \enm

Using modified Jacobi triple product identity~\eqref{modi-Jacobi},
the last two infinite products have the
following series representation
\[\sum_{i,j=0}^\infty (-1)^{i+j}q^{\zbnm{i+1}{2}+3\zbnm{j+1}{2}}
      \{\zeta^{2i}+\zeta^{2i-2}+\cdots+\zeta^{-2i}\}
      \{\zeta^{2j}+\zeta^{2j-2}+\cdots+\zeta^{-2j}\}.\]
Observe the congruence relation
\[\binm{i+1}{2}+3\binm{j+1}{2}+3
\equiv8\Big\{\binm{i+1}{2}+3\binm{j+1}{2}+3\Big\}
\equiv (2i+1)^2+3(2j+1)^2\equiv0\pmod5,\]
which can be reached only if
$i\equiv2\pmod5$ and $j\equiv2\pmod5$
since the corresponding residues modulo $5$ read respectively as
\[(2i+1)^2\equiv0,1,4\pmod5,\quad\text{and}\quad3(2j+1)^2\equiv0,2,3\pmod5.\]
When $i\equiv2\pmod5$ and $j\equiv2\pmod5$, we have
$\{\zeta^{2i}+\zeta^{2i-2}+\cdots+\zeta^{-2i}\}
      \{\zeta^{2j}+\zeta^{2j-2}+\cdots+\zeta^{-2j}\}=0$.
We see that in the $q$-expansion on the right side of the last
equation the coefficient of $q^n$ is zero when $n\equiv2\pmod5$.
The proof of Theorem~\ref{c} has been finished.
\end{proof}

Let
\[C_{2^23^2}=\{(2\lam_1,2\lam_2,3\lam_3,3\lam_4)|
\lam_1,\lam_2,\lam_3,\lam_4\in\mc{P}\}.\] For $\lam\in
C_{\!2^2\!3^2\!}$, we define the sum of parts $s_{\!2^2\!3^2\!}(\!\lam\!)$,
and rank analog
$r_{\!2^2\!3^2\!}(\!\lam)$ by
\bnm
s_{\!2^2\!3^2\!}(\lam)&=&2\sigma(\lam_1)+2\sigma(\lam_2)+3\sigma(\lam_3)+3\sigma(\lam_4)\\
r_{2^23^2}(\lam)&=&\#(\lam_1)-\#(\lam_2)+\#(\lam_3)-\#(\lam_4).
\enm
Define $N_{\!C_{\!2^2\!3^2}\!}(m,n)$ as
the number of 4-colored partitions of $n$ if $s_{\!2^2\!3^2\!}(\lam)\!=\!n$
having $r_{\!2^2\!3^2\!}(\!\lam)\!=\!m$, and $N_{\!C_{2^2\!3^2}\!}(m,t,n)$
as the number of such 4-colored partitions of $n$ having rank analog
$r_{\!2^2\!3^2\!}(\lam)\equiv m\pmod t$. Now, summing over all
4-colored partitions $\!\lam\in C_{2^2\!3^2}$ gives
\[N_{\!C_{\!2^2\!3^2}\!}(m,n)
\!=\!\sum_{\double{\lam\in
C_{2^23^2},s_{\!2^2\!3^2\!}(\lam)=n,}{r_{2^23^2}(\lam)=m}}1.\]

Since
\[r_{2^23^2}(\lam_1,\lam_2,\lam_3,\lam_4)
=-r_{2^23^2}(\lam_2,\lam_1,\lam_4,\lam_3),\] hence
\[N_{\!C_{2^23^2}\!}(m,n)
\!=\!N_{\!C_{2^23^2}\!}(\!-\!m,n) \quad\text{and}\quad
N_{\!C_{2^23^2}\!}(m,t,n)
\!=\!N_{\!C_{2^23^2}\!}(t\!-\!m,t,n).\]
Then we have
\eqn{\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{C_{2^23^2}}(m,n)z^mq^n
=\label{NC2^23^2gf}
\frac{1}
{(zq^2;q^2)_\infty(z^{-1}q^2;q^2)_\infty(zq^3;q^3)_\infty(z^{-1}q^3;q^3)_\infty}.}
By putting $z=1$ in the identity~\eqref{NC2^23^2gf}, we find
\[\sum_{m=-\infty}^{\infty}N_{C_{2^23^2}}(m,n)=\rho(n),\]
where $\rho(n)$ is defined by
$\sum_{n=0}^\infty \rho(n)q^n=\frac1{(q^2;q^2)^2_\infty(q^3;q^3)^2_\infty}$.

\begin{theorem}\label{rho}
For $n\geq0$,
\[N_{\!C_{2^23^2}\!}(0,5,5n+k)\!=\!N_{\!C_{2^23^2}\!}(1,5,5n+k)
\!=\!N_{\!C_{2^23^2}\!}(2,5,5n+k)
\!=\!\frac{\rho(5n\!+\!k)}5; \:k=1,4.\]
\end{theorem}
It can also prove the identity in Zhang and Wang~\cite{zhang-wang}:
$\rho(5n+1)\equiv0\pmod5$ and $\rho(5n+4)\equiv0\pmod5$.
\begin{proof}
The proof of Theorem~\ref{rho} is similar to Theorem~\ref{c}.
Replacing $z$ by $\zeta$
in \eqref{NC2^23^2gf},
we get
\bnm
\+\+\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{C_{2^23^2}}(m,n)\zeta^mq^n
=\frac{1}{\ffat{\zeta q^2,\zeta^{-1}q^2}{q^2}{\infty}
      \ffat{\zeta q^3,\zeta^{-1}q^3}{q^3}{\infty}}\\
\+=\+\frac{1}
        {(q^{10};q^{10})_\infty(q^{15};q^{15})_\infty}
        \times\ffat{q^2,\zeta^2 q^2, \zeta^{-2}q^2}{q^2}{\infty}
            \ffat{q^3,\zeta^2q^3,\zeta^{-2}q^3}{q^3}{\infty}. \enm

Applying modified Jacobi triple product identity \eqref{modi-Jacobi},
we transform the last two infinite products as follows
\[\sum_{i,j=0}^\infty (-1)^{i+j}q^{2\zbnm{i+1}{2}+3\zbnm{j+1}{2}}
      \{\zeta^{2i}+\zeta^{2i-2}+\cdots+\zeta^{-2i}\}
      \{\zeta^{2j}+\zeta^{2j-2}+\cdots+\zeta^{-2j}\}.\]
It is not hard to check that
the residues of q-exponent in the formal power series just displayed
$2\binm{i+1}{2}+3\binm{j+1}{2}$ modulo $5$ are given by the following table:
\centro{\begin{tabular}{||c|c|c|c|c|c||} \hline
\hline
$j\backslash i$&$0$&$1$&$2$&$3$&$4$\\
\hline
$0$&$0$&$2$&$1$&$2$&$0$\\
\hline
$1$&$3$&$0$&$4$&$0$&$3$\\
\hline
$2$&$4$&$1$&$0$&$1$&$4$\\
\hline
$3$&$3$&$0$&$4$&$0$&$3$\\
\hline
$4$&$0$&$2$&$1$&$2$&$0$\\
\hline \hline
\end{tabular}}

When $i\equiv2\pmod5$ or $j\equiv2\pmod5$, we have
$\{\zeta^{2i}+\zeta^{2i-2}+\cdots+\zeta^{-2i}\}
      \{\zeta^{2j}+\zeta^{2j-2}+\cdots+\zeta^{-2j}\}=0$.
We observe that in the $q$-expansion on the right side of the last
equation the coefficient of $q^n$ is zero when $n\equiv1\pmod5$
and $n\equiv4\pmod5$. The
proof of Theorem~\ref{rho} has been completed.
\end{proof}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Rank and crank analogs for colored partitions congruences modulo 7}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
In this section, we define statistics that divide the relevant
partitions into equinumerous classes and provide the combinatorial
interpretation according to~\cite{kim11}
for colored partitions congruences modulo 7 given in
Toh~\cite{toh}, Zhang and Wang~\cite{zhang-wang}.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

If we denote
\[C_{1^32^{-2}}
=\{(2\lam_1,2\lam_2,\lam_3,\lam_4,\lam_5)|\lam_1,\lam_2\in\mc{P},
\lam_3,\lam_4,\lam_5\in\mc{P^*}\},\] then we can call
them as partitions into 5-colors.

For the set of the colored partitions, we define
the sum of parts $s_{1^32^{-2}}(\lam)$, a weight $wt_{1^32^{-2}}(\lam)$ and a
crank analog $c_{1^32^{-2}}(\lam)$ by
\bnm
s_{1^32^{-2}}(\lam)
&=&2\sigma(\lam_1)+2\sigma(\lam_2)+\sigma^*(\lam_3)
+\sigma^*(\lam_4)+\sigma^*(\lam_5)\\
wt_{1^32^{-2}}(\lam)&=&(-1)^{\#(\lam_1)+\#(\lam_2)}
wt(\lam_3)wt(\lam_4)wt(\lam_5)\\
c_{1^32^{-2}}(\lam)&=&c^*(\lam_3)+2c^*(\lam_4)+3c^*(\lam_5),
\enm
where the definitions of $\mc{P^*}$, $\sigma^*(\lam)$, $wt(\lam)$
and $c^*(\lam)$ are presented in section 2.
Let
$M_{\!C_{\!1^3\!2^{\!-\!2}}\!}(m,n)$ denote the number of
5-colored partitions of $n$ if
$s_{\!1^3\!2^{\!-\!2}}(\!\lam)\!=\!n$ (counted according
to the weight $wt_{\!1^3\!2^{\!-\!2}}(\lam)$) with analog of crank
$c_{1^32^{-2}}(\lam)=m$, and
$M_{\!C_{\!1^3\!2^{\!-\!2}}\!}(m,t,n)$ denote the number of
5-colored partitions of $n$ with analog of crank $c_{1^32^{-2}}(\lam)$
congruent to $m \pmod t$,
so that
\[M_{C_{1^32^{-2}}}(m,n)
=\sum_{\double{\lam\in
C_{\!1^3\!2^{\!-\!2}},s_{\!1^3\!2^{\!-\!2}}(\lam)=n,}{c_{1^32^{-2}}(\lam)=m}}
 wt_{1^32^{-2}}(\lam).\]
Then we have
\eqn{\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^32^{-2}}}(m,n)z^mq^n
=\label{MC1^32^-2gf}
\frac{(q^2;q^2)^2_\infty(q;q)^3_\infty}
{\ffat{zq,z^{-1}q,z^2q,z^{-2}q,z^{3}q,z^{-3}q}{q}{\infty}}.}
By putting $z=1$ in the identity~\eqref{MC1^32^-2gf}, we find
\[\sum_{m=-\infty}^{\infty}M_{C_{1^32^{-2}}}(m,n)
=Q_{(po,\overline{p})}(n),\]
where $Q_{(po,\overline{p})}(n)$ is defined by
$\sum_{n=0}^\infty Q_{(po,\overline{p})}(n)q^n
:=\frac{(-q;q)_\infty}{(q;q^2)_\infty(q;q)_\infty}
=\frac{(q^2;q^2)^2_\infty}{(q;q)^3_\infty}$.

Suppose $\varpi$ is primitive seventh root of unity. By letting $z=\varpi$
in \eqref{MC1^32^-2gf}, we have
\bnm
\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{\!1^3\!2^{\!-\!2}}}(m,n)\varpi^mq^n
\!=\!\frac{(q^2;q^2)^2_\infty(q;q)^3_\infty}
{\ffat{\varpi q,\varpi^{\!-\!1}q,\varpi^2q,\varpi^{\!-\!2}q,
\varpi^{3}q,\varpi^{\!-\!3}q}{q}{\infty}}
\!=\!\frac{(q^2;q^2)^2_\infty(q;q)^4_\infty}{(q^7;q^7)_\infty}.
\enm
Utilizing product identity
\eqref{modify-winqq}, we compute the numerator of
the right hand side of last identity as follows
\eqn{\xxqdn
\sum_{i,j={-\infty}}^{+\infty}
(-1)^{i+j}q^{6\zbnm{i}{2}+6\zbnm{j}{2}+3i-j+1}
(2+3i-3j)(3i+3j-2)\label{1422}.}
We illustrate that
the residues of q-exponent in the formal power series just displayed
$6\binm{i}{2}+6\binm{j}{2}+3i-j+1$ modulo $7$ are given by the following table:
\centro{\begin{tabular}{||c|c|c|c|c|c|c|c||} \hline
\hline
$j\backslash i$&$0$&$1$&$2$&$3$&$4$&$5$&$6$\\
\hline
$0$&$1$&$4$&$6$&$0$&$0$&$6$&$4$\\
\hline
$1$&$0$&$3$&$5$&$6$&$6$&$5$&$3$\\
\hline
$2$&$5$&$1$&$3$&$4$&$4$&$3$&$1$\\
\hline
$3$&$2$&$5$&$0$&$1$&$1$&$0$&$5$\\
\hline
$4$&$5$&$1$&$3$&$4$&$4$&$3$&$1$\\
\hline
$5$&$0$&$3$&$5$&$6$&$6$&$5$&$3$\\
\hline
$6$&$1$&$4$&$6$&$0$&$0$&$6$&$4$\\
\hline \hline
\end{tabular}}
The power of $q$ is congruent to $2$ modulo $7$ only when
$i\equiv_70$ and $j\equiv_73$.
Since the coefficient of $q^n$ on the right side of the last
identity is a multiple of $7$ when $n\equiv2\pmod7$,
and $1+\varpi+\varpi^2+\varpi^3+\varpi^4+\varpi^5+\varpi^6$ is a
minimal polynomial in $\mf{Z}[\varpi]$, we must have the result
following as
\begin{theorem}
For $n\geq0$ and $0\leq i< j\leq 6$, we have
%\bnm
%&&M_{\!C_{1^32^{\!-2}}\!}(0,7,7n+2)\!\equiv\!M_{\!C_{1^32^{\!-2}}\!}(1,7,7n+2)
%\!\equiv\!M_{\!C_{1^32^{\!-2}}\!}(2,7,7n+2)\\
%&\equiv&M_{\!C_{1^32^{\!-2}}\!}(3,7,7n+2)\equiv\cdots\equiv
%M_{\!C_{1^32^{\!-2}}\!}(6,7,7n+2)
%\!\equiv\!0\pmod7.
%\enm
\[M_{\!C_{1^32^{\!-2}}\!}(i,7,7n+2)
\!\equiv\!M_{\!C_{1^32^{\!-2}}\!}(j,7,7n+2)\pmod7.\]
\end{theorem}
It can also prove the identity in Toh~\cite{toh}:
$Q_{(po,\overline{p})}(7n+2)\equiv0\pmod7$.

Next we define
\[C_{1^44^42^{\!-\!7}}\!=\!\{\!(\!\lam_1,\lam_2,\lam_3,\lam_4,2\!\lam_5
,2\!\lam_6,2\!\lam_7,2\!\lam_8,2\!\lam_9\!)|\lam_1\!,\lam_2\!,\lam_3\!,\lam_4\!\in\mc{D}\mc{O},
\lam_5\!,\lam_6\!\in\mc{P},\lam_7\!,\lam_8\!,\lam_9\!\in\mc{P^*}\!\}.\]
For $\lam\!=\!(\!\lam_1\!,\lam_2\!,\lam_3\!,\lam_4\!,2\!\lam_5\!
,2\!\lam_6\!,2\!\lam_7\!,2\!\lam_8\!,2\!\lam_9\!)$, we
denote the sum of parts $s_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)$,
a weight $wt_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)$ and a crank analog
$c_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)$ by
\bnm
s_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)
&=&\sigma(\!\lam_1\!)\!+\!\sigma(\!\lam_2\!)
\!+\!\sigma(\!\lam_3\!)\!+\!\sigma(\!\lam_4\!)
\!+\!2\sigma(\!\lam_5\!)\!+\!2\sigma(\!\lam_6\!)
\!+\!2\sigma^*(\!\lam_7\!)\!+\!2\sigma^*(\!\lam_8\!)\!+\!2\sigma^*(\!\lam_9\!)\\
wt_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)
&=&(-1)^{\#(\lam_5)+\#(\lam_6)}
wt(\lam_7)wt(\lam_8)wt(\lam_9)\\
c_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)
&=&c^*(\lam_7)+2c^*(\lam_8)+3c^*(\lam_9). \enm
Finally define
$M_{\!C_{\!1^4\!4^4\!2^{\!-\!7}\!}\!}(m,n)$ as the number of
9-colored partitions of $n$ if
$s_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)\!=\!n$
with crank analog
$c_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)=m$
counted according to the weight $wt_{\!1^4\!4^4\!2^{\!-\!7}\!}(\lam)$ as follows:,
\[M_{C_{\!1^4\!4^4\!2^{\!-\!7}\!}}(m,n)
=\sum_{\double{\lam\in
C_{\!1^4\!4^4\!2^{\!-\!7}\!},s_{\!1^4\!4^4\!2^{\!-\!7}\!}(\lam)=n,}
{c_{\!1^4\!4^4\!2^{\!-\!7}\!}(\lam)=m}}
 wt_{\!1^4\!4^4\!2^{\!-\!7}\!}(\lam).\]
Let $M_{\!C_{\!1^4\!4^4\!2^{\!-\!7}\!}\!}(m,t,n)$ denote the number of
9-colored partitions of $n$ with
crank analog $c_{\!1^4\!4^4\!2^{\!-\!7}\!}(\!\lam\!)$
congruent to $m \pmod t$.

Then we have
\eqn{
\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{\!C_{\!1^4\!4^4\!2^{\!-\!7}}\!}(m,n)z^mq^n
=\label{MC1^44^42^-7gf}
\frac{(-q;q^2)_\infty^4(q^2;q^2)_\infty^5}
{\ffat{zq^2,z^{\!-\!1}q^2,z^{2}q^2,z^{\!-\!2}q^2,
z^{3}q^2,z^{\!-\!3}q^2}{q^2}{\infty}}.}
By replacing $z$ by $1$ in the identity~\eqref{MC1^44^42^-7gf}, we
discover
\[\sum_{m=-\infty}^{\infty}M_{C_{1^44^42^{-7}}}(m,n)=\gamma(n),\]
where $\gamma(n)$ is defined by
$\sum_{n=0}^\infty\gamma(n)q^n=\frac{(-q;q^2)^4_\infty}{(q^2;q^2)_\infty}$.
(see \cite{zhang-wang}).
\begin{theorem}
For $n\geq0$,
\[M_{C_{\!1^44^42^{\!-\!7}}\!}(0,7,7n\!+\!2)
\!\equiv\!M_{\!C_{1^44^42^{\!-\!7}}\!}(1,7,7n\!+\!2)
\equiv\cdots
\!\equiv\!M_{C_{\!1^44^42^{\!-\!7}}\!}(6,7,7n\!+\!2)\pmod7.\]
\end{theorem}
It can also prove the identity
$\gamma(7n+2)\equiv0\pmod7$.
\begin{proof}
Put $z\!=\!\varpi$ in \eqref{MC1^44^42^-7gf} and apply
product identity \eqref{modify-winqq} substituting $q\to-q$ to obtain
\bnm
&&\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^44^42^{-7}}}(m,n)\varpi^mq^n
=\frac{(-q;q^2)_\infty^4(q^2;q^2)_\infty^5}
{\ffat{\varpi q^2,\varpi^{\!-\!1}q^2,\varpi^{2}q^2,\varpi^{\!-\!2}q^2,
\varpi^{3}q^2,\varpi^{\!-\!3}q^2}{q^2}{\infty}}\\
&=&\frac{(-q;q^2)_\infty^4(q^2;q^2)_\infty^6}{(q^{14};q^{14})_\infty}
=\frac{-1}{(q^{14};q^{14})_\infty}
\sum_{i,j={-\infty}}^{+\infty}
q^{6\zbnm{i}{2}+6\zbnm{j}{2}+3i-j+1}
(2+3i-3j)(3i+3j-2).
\enm
We discover that the double sum of the last identity
is similar as~\eqref{1422}. Then we can
use the same congruence relations.
Since the coefficient of $q^n$ on the right side of the last
identity is a multiple of $7$ when $n\equiv2\pmod7$,
and $1+\varpi+\varpi^2+\varpi^3+\varpi^4+\varpi^5+\varpi^6$ is a
minimal polynomial in $\mf{Z}[\varpi]$,
we deduce the theorem.
\end{proof}

If we denote
\[C_{1^22}=\{(\lam_1,\lam_2,2\lam_3,2\lam_4,2\lam_5)|
\lam_1,\lam_2,\lam_3,\lam_4,\lam_5\in\mc{P}\}.\]
It can be said as partitions into
5-colors.
For $\lam=(\lam_1,\lam_2,2\lam_3,2\lam_4,2\lam_5)\in C_{1^22}$, we
define the sum of parts $s_{1^22}(\lam)$, a weight $w_{1^22}(\lam)$ and a
rank analog $r_{1^22}(\lam)$ by
\bnm
s_{1^22}(\lam)&=&\sigma(\lam_1)+\sigma(\lam_2)+2\sigma(\lam_3)
+2\sigma(\lam_4)+2\sigma(\lam_5)\\
w_{1^22}(\lam)&=&(-1)^{\#(\lam_5)}\\
r_{1^22}(\lam)&=&\#(\lam_1)-\#(\lam_2)+3\#(\lam_3)-3\#(\lam_4). \enm
Let
$N_{\!C_{1^2\!2}\!}(m,n)$ denote the number of 5-colored
partitions of $n$ if $s_{1^2\!2}(\lam)=n$ (counted according to the weight
$w_{1^2\!2}(\lam)$) with rank analog $r_{1^22}(\lam)=m$, and
$N_{\!C_{1^2\!2}}(\!m,t,n\!)$ denote the number of
5-colored partitions of $n$ with rank analog $r_{1^22}(\lam)$ congruent to
$m \pmod{\!t}$, hence
\[N_{C_{1^2\!2}\!}(m,n)
=\sum_{\double{\lam\in
C_{1^22},s(\lam)=n,}{r_{1^22}(\lam)=m}}
w_{1^22}(\lam).\]
By considering the transformation that
interchanges $\lam_1$ and $\lam_2$, $\lam_3$ and $\lam_4$, we get
\[N_{C_{1^22}}(m,n)\!=\!N_{C_{1^22}}(-m,n),
\quad N_{C_{1^22}}(m,t,n)\!=\!N_{C_{1^22}}(t-m,t,n).\]
Then we have
\eqn{\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{C_{1^22}}(m,n)z^mq^n
=\label{NC1^22gf}
\frac{(q^2;q^2)_\infty}
{(zq;q)_\infty(z^{-1}q;q)_\infty
(z^3q^2;q^2)_\infty(z^{-3}q^2;q^2)_\infty}.} By
putting $z=1$ in the identity~\eqref{NC1^22gf}, we check
\[\sum_{m=-\infty}^{\infty}N_{C_{1^22}}(m,n)=\alpha(n),\]
where $\alpha(n)$ is defined by
$\sum_{n=0}^\infty\alpha(n)q^n=\frac{1}{(q;q)_\infty^2(q^2;q^2)_\infty}$.
(see~\cite{zhang-wang}).

Suppose $\varpi$ is primitive $7th$ root of unity. Substituting
$z=\varpi$ into \eqref{NC1^22gf},
we have
\bnm \+\+\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{\!C_{\!1^2\!2}\!}(m,n)\varpi^mq^n
=\frac{(q^2;q^2)_\infty}{\ffat{\varpi q,q/\!\varpi}{q}{\infty}
\ffat{\varpi^3 q^2,q^2/\!{\varpi^3}}{q^2}{\infty}} \\
\+=\+\frac{\ffat{q,\varpi^2 q,q/\!{\varpi^2}}{q}{\infty}
          \ffat{q^2,\varpi^3 q,q/\!{\varpi^3}}{q^2}{\infty}
          }
          {(q^7;q^7)_\infty}\\
\+=\+\frac {\sum_{i\geq0}(-1)^i(\varpi^{2i}+\varpi^{2i-2}
                 +\cdots+\varpi^{-2i})q^{\zbnm{i+1}{2}}
\sum_{j=-\infty}^\infty(-1)^j\varpi^{3j}q^{2\zbnm{j}{2}+j}}
      {(q^7;q^7)_\infty}.
\enm
The last line depends only on modified Jacobi identity
\eqref{modi-Jacobi} and classical Jacobi identity~\eqref{jacobi}.

If and only if $i\equiv_73$, we have
$\varpi^{2i}+\varpi^{2i-2}+\cdots+\varpi^{-2i}=0$. Obviously
\begin{equation}\label{1221}
\binm{i+1}{2} \equiv_7\begin{cases}
0,&i\equiv_70,6;\\
1,&i\equiv_71,5;\\
3,&i\equiv_72,4;\\
6,&i\equiv_73;
\end{cases}
\quad\text{and}\quad
2\binm{j}{2}+j \equiv_7\begin{cases}
0,&j\equiv_70;\\
1,&j\equiv_71,6;\\
4,&j\equiv_72,5;\\
2,&m\equiv_73,4.
\end{cases}
\end{equation}
The power of $q$ is congruent to 6 modulo 7 only when
$\binm{i+1}{2}\equiv_76$
and $2\binm{j}{2}+j\equiv_70$ in which
case $i\equiv_73$ and $j\equiv_70$ and the coefficient of $q^{7n+6}$
in the last identity is zero. Since
$1+\varpi+\varpi^2+\varpi^3+\varpi^4+\varpi^5+\varpi^6$ is a minimal polynomial in
$\mf{Z}[\varpi]$, our main result is as follows.
\begin{theorem}~\label{alpha}
For $n\geq0$,
\[N_{C_{1^2\!2}}(0,7,7n\!+\!6)\!
=\!N_{C_{1^2\!2}}(1,7,7n\!+\!6)
\!=\!N_{C_{1^2\!6}}(2,7,7n\!+\!6)\!
=\!\cdots
=\!N_{C_{1^2\!6}}(6,7,7n\!+\!6)\!
=\!\frac{\alpha(7n\!+\!6)}7.\]
\end{theorem}
It can also prove the identity
$\alpha(7n+6)\equiv0\pmod7$.

Denote
\[C_{1^54^52^{\!-\!7}}=\{(\lam_1,\lam_2,\lam_3,\lam_4,\lam_5,2\lam_6,2\lam_7,2\lam_8)|
                 \lam_1,\lam_2,\lam_3,\lam_4,\lam_5\in\mc{D}\mc{O},
\lam_6,\lam_7,\lam_8\in\mc{P^*}\}.\]
We call the elements of
$C_{1^54^52^{\!-\!7}}$ 8-colored partitions.
For $\lam\in
C_{1^54^52^{\!-\!7}}$, we define the sum of parts
$s_{1^54^52^{\!-\!7}}(\!\lam\!)$, a weight
$wt_{1^54^52^{\!-\!7}}(\!\lam\!)$ and a crank analog
$c_{1^54^52^{\!-\!7}}(\!\lam\!)$ by
\bnm
s_{1^54^52^{\!-\!7}}(\lam)&=&\sigma(\lam_1)+\sigma(\lam_2)+\sigma(\lam_3)+\sigma(\lam_4)
+\sigma(\lam_5)+2\sigma^*(\lam_6)+2\sigma^*(\lam_7)+2\sigma^*(\lam_8)\\
wt_{1^54^52^{\!-\!7}}(\lam)&=&wt(\lam_6)wt(\lam_7)wt(\lam_8)\\
c_{1^54^52^{\!-\!7}}(\lam)&=&c^*(\lam_6)+2c^*(\lam_7)+3c^*(\lam_8).
\enm
Let $M_{\!C_{\!1^5\!4^5\!2^{\!-\!7}}\!}(m,n)$ denote the number
of 8-colored partitions of $n$ if $s_{1^54^52^{-7}}(\lam)\!=\!n$ (counted according
to the weight $wt_{\!1^5\!4^5\!2^{\!-\!7}\!}(\!\lam\!)$) with
crank analog $c_{1^54^52^{\!-\!7}}(\lam)=m$,
and $M_{\!C_{\!1^5\!4^5\!2^{\!-\!7}}\!}(m,t,n)$ denote the number of
8-colored partitions of $n$ with crank analog
$c_{1^54^52^{\!-\!7}}(\lam)\equiv t\pmod m$, so
that
\[M_{C_{1^54^52^{\!-\!7}}}(m,n)
=\sum_{\double{\lam\in
C_{1^54^52^{\!-\!7}},s_{1^54^52^{\!-\!7}}(\lam)=n,}{c_{1^54^52^{\!-\!7}}(\lam)=m}}
    wt_{1^54^52^{\!-\!7}}(\lam).\]
Then the generating function is
\eqn{\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^54^52^{\!-\!7}}}(m,n)z^mq^n
=\label{MC1^54^52-7gf}
\frac{(-q;q^2)_\infty^5(q^2;q^2)^3_\infty}
{\ffat{zq^2,z^{-1}q^2,z^2q^2,z^{-2}q^2,z^3q^2,z^{-3}q^2}{q^2}{\infty}}.}
By putting $z=1$ in the identity~\eqref{MC1^54^52-7gf},
we discover
\[\sum_{m=-\infty}^{\infty}M_{C_{1^54^52^{\!-\!7}}}(m,n)=\nu(n),\]
where $\nu(n)$ is defined by
$\sum_{n=0}^\infty\nu(n)q^n=\frac{(-q,q^2)^5_\infty}{(q^2;q^2)^3_\infty}$.
\begin{theorem}~\label{nu}
For $n\geq0$,
\[M_{\!C_{\!1^5\!4^5\!2^{\!-\!7}\!}}(0,7,7n\!+\!6)
\!\equiv\!M_{\!C_{\!1^5\!4^5\!2^{\!-\!7}\!}}(1,7,7n\!+\!6)
\equiv\cdots
\!\equiv\!M_{\!C_{\!1^5\!4^5\!2^{\!-\!7}\!}}(6,7,7n\!+\!6)
\pmod7.\]
\end{theorem}
It can also prove the identity
$\nu(7n+6)\equiv0\pmod7$.
\begin{proof}
By replacing $z$ by $\varpi$ in \eqref{MC1^54^52-7gf},
we write
\bnm
\+\+\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}N_{C_{1^34^12^{-3}}}(m,n)\varpi^mq^n
=\frac{(-q;q^2)_\infty^5(q^2;q^2)^3_\infty}
{\ffat{\varpi q^2,\varpi^{-1}q^2,\varpi^2q^2,\varpi^{-2}q^2,
\varpi^3q^2,\varpi^{-3}q^2}{q^2}{\infty}}\\
\+=\+\frac{(-q;q^2)_\infty^5(q^2;q^2)^4_\infty}
        {(q^{14};q^{14})_\infty}. \enm

Consider
\[(q;q)^3_\infty\ffat{q^2,q,q}{q^2}{\infty}
=\sum_{i=0}^\infty\sum_{j=-\infty}^\infty
(-1)^{i+j}(2i+1)q^{\zbnm{i+1}{2}+j^2},\]
which can be deduced by
Jacobi identity \eqref{jacobi} and \eqref{modi-Jacobi}.

Replacing $q$ by $-q$ in the last identity, we have the
following series representation
\[(-q;q^2)^5_\infty(q^2;q^2)^4_{\infty}
=\sum_{i=0}^\infty\sum_{j=-\infty}^\infty
(-1)^{\zbnm{i}{2}}(2i+1)q^{\zbnm{i+1}{2}+j^2}.\]
If and only if $i\equiv_73$, we have
$2i+1\equiv_70$.
We see that in the $q$-expansion on the right side of the last
equation the coefficient of $q^n$ is a multiple of $7$
when $n\equiv6\pmod7$ referring to~\eqref{1221}. The
proof of Theorem~\ref{nu} has been finished.
\end{proof}
Let
\[C_{1^52^3}=\{(\!\lam_1,\!\lam_2,\!\lam_3,\!\lam_4,
\!\lam_5,2\!\lam_6,2\!\lam_7,2\!\lam_8)|
                 \lam_1,\lam_2,\lam_3,\lam_4\in\mc{P},
                 \lam_5,
\lam_6,\lam_7,\lam_8\in\mc{P^*}\}.\] We can say them as
partitions into 8-colors. For $\lam\in C_{1^52^3}$, we denote
the sum of parts $s_{1^52^3}(\lam)$, a weight $wt_{1^52^3}(\lam)$ and a
crank analog $c_{1^52^3}(\lam)$ by \bnm
s_{1^52^3}(\lam)&=&\sigma(\lam_1)+\sigma(\lam_2)+\sigma(\lam_3)+\sigma(\lam_4)
+\sigma^*(\lam_5)+2\sigma^*(\lam_6)+2\sigma^*(\lam_7)+2\sigma^*(\lam_8)\\
wt_{1^52^3}(\lam)&=&wt(\lam_5)wt(\lam_6)wt(\lam_7)wt(\lam_8)\\
c_{1^52^3}(\lam)&=&\#(\lam_1)\!-\!\#(\lam_2)\!+\!2\#(\lam_3)\!-\!2\#(\lam_4)
\!+\!3c^*(\lam_5)\!+\!c^*(\lam_6)\!+\!2c^*(\lam_7)\!+\!3c^*(\lam_8).
\enm
The number of 8-colored partitions of $n$ if $s_{1^52^3}(\lam)\!=\!n$
with crank analog $c_{1^52^3}(\lam)=m$
counted according to the weight $wt_{\!1^5\!2^3\!}(\lam)$
is denoted by
$M_{\!C_{\!1^5\!2^3\!}}(m,n)$, so that
\[M_{C_{1^52^3}}(m,n)
=\sum_{\double{\lam\in
C_{1^52^3},s_{1^52^3}(\lam)=n,}{c_{1^52^3}(\lam)=m}}
    wt_{1^52^3}(\lam).\]
The number of 8-colored partitions of $n$
with crank analog $c_{1^52^3}(\lam)$ congruent
to $m \pmod t$ is denoted by $M_{C_{1^52^3}}(m,t,n)$.
The
following generating function for $M_{C_{1^52^3}}(m,n)$ is
\bmn
&&\nnm\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^52^3}}(m,n)z^mq^n\\
&=&\label{MC1^52^3gf}
\frac{(q;q)_\infty(q^2;q^2)^3_\infty}
{\ffat{zq,z^{\!-\!1}q,z^{2}q,z^{\!-\!2}q,
z^{3}q,z^{\!-\!3}q}{q}{\infty}
\ffat{zq^2,z^{\!-\!1}q^2,z^{2}q^2,z^{\!-\!2}q^2,
z^{3}q^2,z^{\!-\!3}q^2}{q^2}{\infty}}.
\emn
By setting $z=1$ in the identity~\eqref{MC1^52^3gf}, we find
\[\sum_{m=-\infty}^{\infty}M_{C_{1^52^3}}(m,n)=\mu(n),\]
where $\mu(n)$ is defined by
$\sum_{n=0}^\infty\mu(n)q^n
=\frac{1}{(q;q)^5_\infty(q^2;q^2)^3_\infty}$.
\begin{theorem}~\label{mu}
For $n\geq0$,
\[M_{\!C_{1^5\!2^3}\!}(0,7,7n\!+\!6)
\!\equiv\!M_{\!C_{1^5\!2^3}\!}(1,7,7n\!+\!6)
\equiv\cdots
\!\equiv\!M_{\!C_{1^5\!2^3}\!}(6,7,7n\!+\!6)
\pmod7.\]
\end{theorem}
It can also prove the identity
$\mu(7n+6)\equiv0\pmod7$.
\begin{proof}
%The proof of Theorem~\ref{mu} is similar to Theorem~\ref{nu}.
By letting $z=\varpi$ in \eqref{MC1^52^3gf}, we get
\bnm
\+\+\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^52^3}}(m,n)\varpi^mq^n\\
\+=\+\frac{(q;q)_\infty(q^2;q^2)^3_\infty}
{\ffat{\varpi q,\varpi^{\!-\!1}q,\varpi^{2}q,\varpi^{\!-\!2}q,
\varpi^{3}q,\varpi^{\!-\!3}q}{q}{\infty}
\ffat{\varpi q^2,\varpi^{\!-\!1}q^2,\varpi^{2}q^2,\varpi^{\!-\!2}q^2,
\varpi^{3}q^2,\varpi^{\!-\!3}q^2}{q^2}{\infty}}\\
\+=\+\frac{(q;q)^2_\infty(q^2;q^2)^4_\infty}
        {(q^7;q^{7})_\infty(q^{14};q^{14})_\infty}.
\enm

Investigating product identity~\eqref{winq-1},
splitting the bilateral sum with respect to $i$
into two unilateral sums,
the numerator infinite products on the last line
of the last formula have the following series expression
\bmn
(q;q)_{\infty}^2(q^2;q^2)_{\infty}^4
=\label{1523}\sum_{i=0}^{\infty}\sum_{j={-\infty}}^{+\infty}
q^{3\zbnm{i+1}{2}+3\zbnm{j}{2}+j}
\frac{(1+3i+3j)(2+3i-3j)}{2}. \emn

We check that
the residues of q-exponent in the formal power series displayed
$3\binm{i+1}{2}+3\binm{j}{2}+j$ modulo $7$ are presented by the following table:
\centro{\begin{tabular}{||c|c|c|c|c|c|c|c||} \hline
\hline
$j\backslash i$&$0$&$1$&$2$&$3$&$4$&$5$&$6$\\
\hline
$0$&$0$&$3$&$2$&$4$&$2$&$3$&$0$\\
\hline
$1$&$1$&$4$&$3$&$5$&$3$&$4$&$1$\\
\hline
$2$&$5$&$1$&$0$&$2$&$0$&$1$&$5$\\
\hline
$3$&$5$&$1$&$0$&$2$&$0$&$1$&$5$\\
\hline
$4$&$1$&$4$&$3$&$5$&$3$&$4$&$1$\\
\hline
$5$&$0$&$3$&$2$&$4$&$2$&$3$&$0$\\
\hline
$6$&$2$&$5$&$4$&$6$&$4$&$5$&$2$\\
\hline \hline
\end{tabular}}
If and only if $3\binm{i+1}{2}+3\binm{j}{2}+j\equiv6\pmod7$,
we have $i\equiv_73$ and $j\equiv_76$.
Since the coefficient of $q^n$ on the right side of the last
identity is a multiple of $7$ when $n\equiv6\pmod7$,
and $1+\varpi+\varpi^2+\varpi^3+\varpi^4+\varpi^5+\varpi^6$ is a
minimal polynomial in $\mf{Z}[\varpi]$,
we finish the proof of Theorem~\ref{mu}.
\end{proof}
Consider
\[C_{1^24^22^{\!-\!3}}=\{(\lam_1,\lam_2,2\lam_3,2\lam_4,2\lam_5,2\lam_6,2\lam_7)|
                 \lam_1,\lam_2\in\mc{D}\mc{O},
                 \lam_3,\lam_4\in\mc{P},
                 \lam_5,\lam_6,\lam_7\in\mc{P^*},\}.\]
We call them as partitions into 7-colors. For $\lam\in
C_{1^24^22^{\!-\!3}}$, we define the sum of parts $s_{1^24^22^{\!-\!3}}(\lam)$, a
weight $wt_{1^24^22^{\!-\!3}}(\lam)$ and a crank analog
$c_{1^24^22^{\!-\!3}}(\lam)$ by
\bnm
s_{1^24^22^{\!-\!3}}(\lam)&=&\sigma(\lam_1)+\sigma(\lam_2)+2\sigma(\lam_3)+2\sigma(\lam_4)
+2\sigma^*(\lam_5)+2\sigma^*(\lam_6)+2\sigma^*(\lam_7)\\
wt_{1^24^22^{\!-\!3}}(\lam)
&=&(-1)^{\#(\lam_3)+\#(\lam_4)}
   wt(\lam_5)wt(\lam_6)wt(\lam_7)\\
c_{1^24^22^{\!-\!3}}(\lam)&=&c^*(\lam_5)+2c^*(\lam_6)+3c^*(\lam_7).
\enm
Let $M_{\!C_{1^24^22^{\!-\!3}}\!}(m,n)$ denote the number
of 7-colored partitions of $n$ if $s_{1^24^22^{\!-\!3}}(\lam)\!=\!n$ (counted according
to the weight $wt_{1^24^22^{\!-\!3}}(\lam)$)
with crank analog $c_{1^24^22^{\!-\!3}}(\lam)=m$, so
that
\[M_{C_{1^24^22^{\!-\!3}}}(m,n)
=\sum_{\double{\lam\in
C_{1^24^22^{\!-\!3}},s_{1^24^22^{\!-\!3}}(\lam)=n,}{c_{1^24^22^{\!-\!3}}(\lam)=m}}
    wt_{1^24^22^{\!-\!3}}(\lam).\]
The number of 7-colored partitions of $n$ with crank analog
$c_{1^24^22^{\!-\!3}}(\lam)\equiv
m\pmod t$ is denoted by $M_{\!C_{1^24^22^{\!-\!3}}\!}(m,t,n)$.
Then the two
variable generating function for
$M_{\!C_{1^24^22^{\!-\!3}}\!}(m,n)$ is
\eqn{\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^24^22^{\!-\!3}}}(m,n)z^mq^n
=\label{MC1^24^22-3gf}
\frac{(-q;q^2)^2_\infty(q^2;q^2)_\infty^5}
{\ffat{zq^2,z^{\!-\!1}q^2,z^{2}q^2,z^{\!-\!2}q^2,z^{3}q^2,z^{\!-\!3}q^2}
{q^2}{\infty}}.}
If we simply put $z=1$ in the identity~\eqref{MC1^24^22-3gf}, and
read off the coefficients of like powers of $q$, we find
\[\sum_{m=-\infty}^{\infty}M_{C_{1^24^22^{\!-\!3}}}(m,n)=\beta(n),\]
where $\beta(n)$ is defined by
$\sum_{n=0}^\infty\beta(n)q^n
=\frac{(-q;q^2)^2_\infty}{(q^2;q^2)_\infty}$.

Putting $z=\varpi$ in \eqref{MC1^24^22-3gf} gives
\bnm \+\+\sum_{m\in
\mb{Z}}\sum_{n=0}^{\infty}M_{C_{1^24^22^{\!-\!3}}}(m,n)\varpi^mq^n
=\frac{(-q;q^2)^2_\infty(q^2;q^2)_\infty^5}
{\ffat{\varpi q^2,\varpi^{\!-\!1}q^2,\varpi^{2}q^2,
\varpi^{\!-\!2}q^2,\varpi^{3}q^2,\varpi^{\!-\!3}q^2}
{q^2}{\infty}}\\
\+=\+\frac{(-q;q^2)^2_\infty(q^2;q^2)_\infty^6}
        {(q^{14};q^{14})_\infty}.
\enm

Substituting $q\to-q$ into identity \eqref{1523},
the numerator infinite products have the
following series expression
\[(-q;q^2)_{\infty}^2(q^2;q^2)_{\infty}^6
=\sum_{i=0}^{\infty}\sum_{j={-\infty}}^{+\infty}
(-1)^{\zbnm{i+1}{2}+\zbnm{j+1}{2}}
q^{3\zbnm{i+1}{2}+3\zbnm{j}{2}+j}
\frac{(1+3i+3j)(2+3i-3j)}{2}. \]
It is easy to find that the power of $q$
is congruent to $6$ modulo $7$ if and only if
$i\equiv_73$ and $j\equiv_76$
considering the congruence relations in the proof of
Theorem~\ref{mu}.
Since the
coefficient of $q^n$ on the right side of the last identity is a multiple of $7$
when $n\equiv_76$, and
$1+\varpi+\varpi^2+\varpi^3+\varpi^4+\varpi^5+\varpi^6$ is a
minimal polynomial in $\mf{Z}[\varpi]$, our main result is as follows:
\begin{theorem}
For $n\geq0$ and $0\leq i< j\leq6$, we obtain
\[M_{C_{1^24^22^{\!-\!3}}}(i,7,7n+6)
\equiv M_{C_{1^24^22^{\!-\!3}}}(j,7,7n+6)
\pmod7.\]
\end{theorem}
It can also prove the identity
$\beta(7n+6)\equiv0\pmod7$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\subsection*{Acknowledgements}
%Thanks to Professor Querty for suggesting the proof of
%Lemma~\ref{lem:Technical}.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% \bibliographystyle{plain}
% \bibliography{myBibFile}
% If you use BibTeX to create a bibliography
% then copy and past the contents of your .bbl file into your .tex file



\end{document}
